Author's response to reviews

Title: Effects of smoking on the genetic risk of obesity: the Population Architecture using Genomics and Epidemiology Study

Authors:

Megan D Fesinmeyer (mfesinme@whi.org)
Kari E North (kari_north@unc.edu)
Unlee Lim (ulim@cc.hawaii.edu)
Petr A Buzkova (buzkova@u.washington.edu)
Dana C Crawford (dana.c.crawford@vanderbilt.edu)
Jeffrey Haessler (haessler@WHI.org)
Myron D Gross (gross001@umn.edu)
Jay H Fowke (jay.fowke@Vanderbilt.EDU)
Robert J Goodloe (robert.j.goodloe@Vanderbilt.EDU)
Shelly-Ann Love (smmeade@email.unc.edu)
Misa I Graff (migraff@email.unc.edu)
Christopher S Carlson (ccarlson@fhcrc.org)
Lewis H Kuller (kullerl@edc.pitt.edu)
Tara C Matise (matise@biology.rutgers.edu)
Ching-Ping Hong (hongx003@umn.edu)
Brian E Henderson (behender@usc.edu)
Melissa Allen (Melissa.Allen@Vanderbilt.EDU)
Rebecca R Rohde (rohde@email.unc.edu)
Ping Mayo (ping.c.mayo@Vanderbilt.EDU)
Nathalie C Schnetz-Boutaud (nathalie.c.schnetz-boutaud@vanderbilt.edu)
Kristine R Monroe (kmonroe@usc.edu)
Marylyn D Ritchie (marylyn.ritchie@psu.edu)
Ross L Prentice (rprentic@whi.org)
Lawrence N Kolonel (lkolonel@cc.hawaii.edu)
JoAnn E Manson (jmanson@rics.bwh.harvard.edu)
James S Pankow (Panko001@umn.edu)
Lucia A Hindorff (hindorff@mail.nih.gov)
Nora Franceschini (noraf@unc.edu)
Lynne R Wilkens (lynee@cc.hawaii.edu)
Christopher A Haiman (Christopher.Haiman@med.usc.edu)
Loic Le Marchand (loic@cc.hawaii.wsu)
Ulrike Peters (upeters@fhcrc.org)

Version: 2 Date: 17 September 2012

Author's response to reviews:

missing Tables 1 & 2 have been added to the manuscript main text and the revised version is being uploaded. There is only one supp Table.